Author's response to reviews

Title: Incidence and outcome of newly-diagnosed tuberculosis infections in schizophrenics: a 12-year, nationwide, retrospective longitudinal study

Authors:

Shu-Chen Kuo (ludwigvantw@gmail.com)
Yung-Tai Chen (ytchen0117@gmail.com)
Szu-Yuan Li (sylivghtpe.gov.tw)
Yi-Tzu Lee (s851009@yahoo.com.tw)
Albert C. Yang (accyang@gmail.com)
Te-Li Chen (tecklayyy@gmail.com)
Chia-Jen Liu (chijenliu@gmail.com)
Tzeng-Ji Chen (tjchen@vghtpe.gov.tw)
Ih-Jen Su (suihjen@nhri.org.tw)
Chang-Phone Fung (cpfung@vghtpe.gov.tw)

Version: 4 Date: 20 July 2013

Author's response to reviews:

Dear Editor,

Thanks for your kind consideration of our manuscript and remind. We had added “As the data set used in our study consisted of de-identified secondary data released to the public for research purposes, the study was exempt from full review by the institutional review board.” in the Method section.

Best,
Dr. Te-Li Chen